Author’s response to reviews

Title: Quinolone Prophylaxis for the Prevention of BK Virus Infection in Kidney Transplantation: Study Protocol for a Pilot Randomized Controlled Trial

Authors:

Atul Humar (ahumar@ualberta.ca)
John Gill (jgill@providnecehealth.bc.ca)
Olwyn Johnston (olwyn.johnston@vch.ca)
Dean Fergusson (dafergusson@ohri.ca)
Andrew House (Andrew.House@lhsc.on.ca)
Louise Lebel (llebel@ohri.ca)
Sandra Cockfield (Sandra.cockfield@ualberta.ca)
S. Joseph Kim (Joseph.Kim@uhn.ca)
Jeff Zaltzman (Jeffrey.zaltzman@utoronto.ca)
Marcelo Cantarovich (Marcelo.cantarovich@muhc.mcgill.ca)
Martin Karpinski (MKarpinski@exchange.hsc.mb.ca)
Tim Ramsay (tramsay@ohri.ca)
Greg Knoll (gknoll@toh.on.ca)

Version: 2 Date: 4 April 2013

Author’s response to reviews:

statement of ethical approval has been added to the Methods section of the manuscript